Author's response to reviews

Title: A novel c.-22T>C mutation in GALK1 promoter is associated with elevated galactokinase phenotype

Authors:

Hyung-Doo Park (nayadoo@hanmail.net)
Yoon-Kyoung Kim (ykkim@hanmail.net)
Kyoung U Park (m91w95@snu.ac.kr)
Jin Q Kim (jgkim@snu.ac.kr)
Young-Han Song (ysong@hallym.ac.kr)
Junghan Song (songjhcp@snu.ac.kr)

Version: 2 Date: 8 October 2008

Author's response to reviews:

Oct 07, 2008

Dear Editor,

We wish to submit for publication in the BMC Medical Genetics, a manuscript entitled, "A novel c.-22T>C mutation in GALK1 promoter is associated with elevated galactokinase phenotype", by Park et al.

Your kind consideration of this paper would be greatly appreciated. These data have not been reported elsewhere. This manuscript has not been submitted elsewhere. All authors took part in the work and agree with the contents of the manuscripts.

There were no financial conflicts of interest, including stock ownership and patents.

I do sincerely hope that your warm consideration would be favorable for us so that we would have opportunity to contribute our paper to your prestigious journal. We look forward to hearing from you good news soon.

Thank you in advance!

Sincerely yours,

Junghan Song, M.D., Ph.D.
Department of Laboratory Medicine
Seoul National University Bundang Hospital